Author's response to reviews

Title: Physical activity promotion in the primary care setting in pre- and type 2 diabetes - the Sophia Step Study, an RCT

Authors:

Jenny Rossen (jenny.rossen@shh.se)
Agneta Yngve (agneta.yngve@oru.se)
Maria Hagströmer (maria.hagstromer@ki.se)
Kerstin Brismar (kerstin.brismar@ki.se)
Barbara E Ainsworth (barbara.ainsworth@asu.edu)
Christina Iskull (christina.iskull@sh.se)
Peter Möller (peter.moller@hlamottagningen.se)
Unn-Britt Johhansson (unn-britt.johansson@shh.se)

Version: 2
Date: 12 June 2015

Author's response to reviews: see over
Changes made to the original submission

1. Ethical and Funding Approval Documentation
These documents were sent as email attachments to BMCSeriesEditorial@biomedcentral.com on 12 June 2015.

2. Funding
The study protocol has been peer-reviewed by the funding bodies as well as by the Ethical board.

3. Study status
The study is ongoing, continuously recruiting participants and collecting data.

4. Related Articles
To date no publications have been published or submitted to any journal on the study.

5. Date for registry
The date that the trial was registered is added alongside the trial registration number in the abstract in both the submission system and in the manuscript. The revised manuscript was uploaded in the system on June 12.

Sincerely yours,

Jenny Rossen
PhD student, Project manager of Sophia Step Study
Karolinska Institutet/Sophiahemmet University
Jenny.rossen@shh.se